
****************************************************************
          NARROW SWATH TIMELINE GMNNOT41 executed
          continued from day 338 (2007/12/04)
          holding on until orbit 66010, time  ~13:00
    ****************************************************************
*******************************************************************************
Summary of Anomalies:
station info

HO orbit 66002 EGOI data missing 05-DEC-2007 00:32:01.438 - 05-DEC-2007 00:46:32.319        870.88100 [sec]
BE orbit 66003 EGOI data missing 05-DEC-2007 01:51:29.313 - 05-DEC-2007 02:02:42.853        673.54000 [sec]
HO orbit 66008 EGOI data missing 05-DEC-2007 11:05:10.968 - 05-DEC-2007 11:13:57.523        526.55500 [sec]
HO orbit 66009 EGOI data missing 05-DEC-2007 12:42:26.068 - 05-DEC-2007 12:57:09.790        883.72200 [sec]
MA orbit 66009 EGOI data missing 05-DEC-2007 11:55:05.033 - 05-DEC-2007 11:59:33.742        268.70900 [sec]
HO orbit 66010 EGOI data missing 05-DEC-2007 14:22:22.820 - 05-DEC-2007 14:34:55.513        752.69300 [sec]
MM orbit 66010 EGOI data missing 05-DEC-2007 14:13:23.640 - 05-DEC-2007 14:26:07.264        763.62400 [sec]
MM orbit 66011 EGOI data missing 05-DEC-2007 15:52:50.068 - 05-DEC-2007 16:05:25.667        755.59900 [sec]
SG orbit 66011 EGOI data missing 05-DEC-2007 16:16:37.873 - 05-DEC-2007 16:28:49.136        731.26300 [sec]
MM orbit 66013 EGOI data missing 05-DEC-2007 19:11:10.966 - 05-DEC-2007 19:23:49.756        758.79000 [sec]
MM orbit 66014 EGOI data missing 05-DEC-2007 20:50:37.630 - 05-DEC-2007 21:03:21.353        763.72300 [sec]
MA orbit 66014 EGOI data missing 05-DEC-2007 19:49:48.544 - 05-DEC-2007 20:02:37.437        768.89300 [sec]
HO orbit 66015 EGOI data missing 05-DEC-2007 22:23:26.250 - 05-DEC-2007 22:35:22.554        716.30400 [sec]
MM orbit 66015 EGOI data missing 05-DEC-2007 22:30:45.318 - 05-DEC-2007 22:43:08.909        743.59100 [sec]
MA orbit 66008 EGOI data gap     05-DEC-2007 10:18:25.289 - 05-DEC-2007 10:25:36.464        431.17500 [sec]
MA orbit 66015 EGOI data gap     05-DEC-2007 21:28:58.308 - 05-DEC-2007 21:31:05.452        127.14400 [sec]
MM orbit 66006 EGOI data gap     05-DEC-2007 07:38:15.288 - 05-DEC-2007 07:40:52.393        157.10500 [sec]
BE orbit 66011 EGOI data gap     05-DEC-2007 14:50:47.974 - 05-DEC-2007 14:59:56.521        548.54700 [sec]

instrument info
EGOI

1 - solar calibration measurements not available due to ERS-2 manoeuvre

*******************************************************************************

-------------------------------------------------------------------------------
   GOME Daily Reports Analysis    05 Dec 2007
-------------------------------------------------------------------------------

Station ID   see above
MPH Product Confidence            OK
SPH Window Information            OK
Command Word Echo Summary         OK
Instrument Status 1A              OK
Instrument Status 1B              OK
Instrument Status 2               OK
Integration Times Channel 1       OK
Co-Adding and Cluster Mode Flags  OK
Integration Times Band 2A         OK
Integration Times Band 2B         OK
Integration Times Band 3          OK
Integration Times Band 4          OK
Scan Mirror Position              >> timeline GMNNOT41 executed as planned,
                                      until Orb. 66010, ~13:00
Polarisation Detectors            OK
FPA Temperatures A                OK
FPA Temperatures B                OK
Charge Amp Temperatures           >> values still out of range due to anomaly of 04/12/07
Other Temperatures A              >> values still out of range due to anomaly of 04/12/07 
DDHU Temperatures                 >> values still out of range due to anomaly of 04/12/07 
Optical Bench Temperatures        >>   
Other Temperatures B              >>    
Calibr. Lamp and Instr. Status 3  OK
Scan Mirror Motor Current         >> pattern not repeated due to execution of
                                     timeline GMNNOT4
Selected Temperature A            >> values still out of range due to anomaly of 04/12/07
Selected Temperature B            >> values still out of range due to anomaly of 04/12/07  
Selected Temperature C            >> values still out of range due to anomaly of 04/12/07  
Channel 1 Summation               OK



Channel 2 Summation               OK
Channel 4 Summation               OK
Log pages                         OK
331/318 nm Uncal. Line Ratio      OK
Uncal. PMDs as RGB signal         OK
780 nm Uncal. Intensity           OK 
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